\documentclass{article}
\usepackage{fullpage}
\usepackage{parskip}
\usepackage{titlesec}
\usepackage{xcolor}
\usepackage[colorlinks = true,
            linkcolor = blue,
            urlcolor  = blue,
            citecolor = blue,
            anchorcolor = blue]{hyperref}
\usepackage[natbibapa]{apacite}
\usepackage{eso-pic}
\AddToShipoutPictureBG{\AtPageLowerLeft{\includegraphics[scale=0.7]{powered-by-Authorea-watermark.png}}}

\renewenvironment{abstract}
  {{\bfseries\noindent{\abstractname}\par\nobreak}\footnotesize}
  {\bigskip}

\titlespacing{\section}{0pt}{*3}{*1}
\titlespacing{\subsection}{0pt}{*2}{*0.5}
\titlespacing{\subsubsection}{0pt}{*1.5}{0pt}

\usepackage{authblk}

\usepackage{graphicx}
\usepackage[space]{grffile}
\usepackage{latexsym}
\usepackage{textcomp}
\usepackage{longtable}
\usepackage{tabulary}
\usepackage{booktabs,array,multirow}
\usepackage{amsfonts,amsmath,amssymb}
\providecommand\citet{\cite}
\providecommand\citep{\cite}
\providecommand\citealt{\cite}
% You can conditionalize code for latexml or normal latex using this.
\newif\iflatexml\latexmlfalse
\providecommand{\tightlist}{\setlength{\itemsep}{0pt}\setlength{\parskip}{0pt}}%

\AtBeginDocument{\DeclareGraphicsExtensions{.pdf,.PDF,.eps,.EPS,.png,.PNG,.tif,.TIF,.jpg,.JPG,.jpeg,.JPEG}}

\usepackage[utf8]{inputenc}
\usepackage[ngerman,english]{babel}



\begin{document}

\title{10 Famous Articles Still Behind a Paywall\\}


 \author{The Authorea Team}


\date{}

\begingroup
\let\center\flushleft
\let\endcenter\endflushleft
\maketitle
\endgroup




Some research findings have permeated all parts of our culture -- so
much so that we often take the actual ideas and \emph{\textbf{access}}
to them for granted. Would you believe that Einstein's "\emph{Does the
Inertia of a Body Depend Upon Its Energy Content?"} remains behind a
\href{http://dx.doi.org/10.1002/andp.200590007}{paywall}, despite being
more than 100 years old?\\

Here we show 10 famous papers that have achieved a similar ``rockstar''
status in the public imagination yet are still subject to paywalls.~At
Authorea, we believe that open communication is the key to advancing
research. We're working every day to make it possible for researchers to
have full control over their work and its dissemination.~\\\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/darwin/darwin}
\caption{{\textbf{On the Tendency of Species to form Varieties; and on the
Perpetuation of Varieties and Species by Natural Means of Selection
(1858)}
\emph{\emph{by C. Darwin Esq., F.R.S., F.L.S., \&c F.G.S., A. Wallace
Esq.~}}
\emph{\emph{\emph{``This MS. work was never intended for publication,
and therefore was not written with care.''}}}
This paper was the first announcement of the groundbreaking theory of
natural selection, which created a new generation of thinking in
biology. Darwin went on to write his famous~\emph{On The Origin of
Species} after this publication. (\protect\cite{Darwin_1858})
\textbf{Citations:~}611
\textbf{Cost:}
\href{http://onlinelibrary.wiley.com/doi/10.1111/j.1096-3642.1858.tb02500.x/pdf}{\$38.00}%
}}
\end{center}
\end{figure}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/DH-Nature/DH-Nature}
\caption{{\hypertarget{auto-label-caption-858694}{}
\textbf{Molecular Structure of Nucleic Acids: A Structure for
Deoxyribose Nucleic Acid (1953)}
\hypertarget{auto-label-caption-269937}{}
\emph{\emph{by J.D. Watson \& F. H. C. Crick~}}
\hypertarget{auto-label-caption-581135}{}
This revolutionary paper proposed the structure of DNA, including how
base pairs, Chargaff ratios, and the two chain structure worked.
(\protect\cite{WATSON_1953})
\hypertarget{auto-label-caption-582859}{}
\textbf{Citations:~}12046
\hypertarget{auto-label-caption-369053}{}
\textbf{Cost:~}\href{http://www.nature.com/nature/journal/v171/n4356/pdf/171737a0.pdf}{\$32.00}
\hypertarget{auto-label-caption-514773}{}
\hypertarget{auto-label-caption-301276}{}%
}}
\end{center}
\end{figure}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/franklin/franklin}
\caption{{\hypertarget{auto-label-caption-847065}{}
\textbf{Molecular Configuration in Sodium Thymonucleate (1953)}
\hypertarget{auto-label-caption-843248}{}
\emph{\emph{by R. E. Franklin \& R.G. Gosling}}
\hypertarget{auto-label-caption-991090}{}
In this paper, Franklin and Gosling suggest that DNA is helical, with
the phosphate groups on the outside of the two strands. This publication
came after the famous Watson \& Crick paper and expanded on these
scientists' work. (\protect\cite{FRANKLIN_1953})
\hypertarget{auto-label-caption-165308}{}
\textbf{Citations:} 1015
\hypertarget{auto-label-caption-863532}{}
\textbf{Cost:}
\href{http://www.nature.com/nature/journal/v171/n4356/pdf/171740a0.pdf}{\$32.00}
\hypertarget{auto-label-caption-689795}{}
\hypertarget{auto-label-caption-278849}{}
\hypertarget{auto-label-caption-858431}{}%
}}
\end{center}
\end{figure}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/95-3/Feynman}
\caption{{\hypertarget{auto-label-caption-885331}{}
\textbf{The Theory of Positrons (1949)}
\hypertarget{auto-label-caption-601751}{}
\emph{\emph{by R.P. Feynman~}}
\hypertarget{auto-label-caption-470636}{}
In this paper, Feynman proposed his interpretation of the positron as an
electron moving backward in time. (\protect\cite{Feynman_1949})
\hypertarget{auto-label-caption-329538}{}
\textbf{Citations:~}1015
\hypertarget{auto-label-caption-233766}{}
\textbf{Cost:~}\href{https://journals.aps.org/cart}{\$25}or
\href{http://www.atticusrarebooks.com/pages/books/95/richard-feynman/space-time-approach-to-quantum-electrodynamics-with-the-theory-of-positrons-in-physical-review}{\$2,750
~}
\hypertarget{auto-label-caption-668311}{}
\hypertarget{auto-label-caption-107199}{}%
}}
\end{center}
\end{figure}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/fx1-lrg1/Warren}
\caption{{\hypertarget{auto-label-caption-928332}{}
\textbf{Unidentified Curved Bacilli on Gastric Epithelium in Active
Chronic Gastritis (1983)}
\hypertarget{auto-label-caption-723289}{}
\emph{\emph{by~J Robin Warren \& Barry Marshall~}}
\hypertarget{auto-label-caption-114532}{}
This paper uncovered a new species of bacteria in the genus
Campylobacter, which proved to be an important factor in diseases like
gastritis. (\protect\cite{ROBINWARREN_1983})
\hypertarget{auto-label-caption-756370}{}
\textbf{Citations:} 4347
\hypertarget{auto-label-caption-115807}{}
\textbf{Cost:~}\href{https://s100.copyright.com/AppDispatchServlet\#formTop}{\$55.20}
\hypertarget{auto-label-caption-506345}{}
\hypertarget{auto-label-caption-239856}{}%
}}
\end{center}
\end{figure}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/Screen-Shot-2016-08-22-at-9.22.47-AM/Dyson}
\caption{{\hypertarget{auto-label-caption-934602}{}
Video:~\href{http://www.webofstories.com/play/freeman.dyson/80}{The
S-matrix paper that made me famous}~
\hypertarget{auto-label-caption-901940}{}
\hypertarget{auto-label-caption-739161}{}
\textbf{S-Matrix in Quantum Electrodynamics (1949)}
\hypertarget{auto-label-caption-793211}{}
\emph{\emph{by F.J. Dyson}}
\hypertarget{auto-label-caption-858089}{}
This paper made big strides in the field of quantum electrodynamics,
formulating some of the most important theories in the field today.
(\protect\cite{Dyson_1949})
\hypertarget{auto-label-caption-137233}{}
\textbf{Citations:~}1532
\hypertarget{auto-label-caption-225519}{}
\textbf{Cost:~}\href{http://journals.aps.org/pr/abstract/10.1103/PhysRev.75.1736}{\$25}
\hypertarget{auto-label-caption-728032}{}
\hypertarget{auto-label-caption-127542}{}%
}}
\end{center}
\end{figure}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/schrodingers-cat/schrodingers-cat}
\caption{\selectlanguage{ngerman}{\hypertarget{auto-label-caption-918366}{}
\textbf{Die gegenwärtige Situation in der Quantenmechanik (The present
situation in quantum mechanics)}~\textbf{(1935)}
\hypertarget{auto-label-caption-807738}{}
\emph{by Erwin Schrödinger}
\hypertarget{auto-label-caption-850210}{}
The paper in which Schrödinger provides a thought experiment in quantum
mechanics: a scenario in which a cat that may be simultaneously both
alive and dead, as a result of being linked to a random subatomic event
that may or may not occur. \emph{} (\emph{\protect\cite{Schr_dinger_1935}})
\hypertarget{auto-label-caption-581689}{}
\textbf{Citations}: 3354
\hypertarget{auto-label-caption-963755}{}
\textbf{Cost}:~\href{http://link.springer.com/article/10.1007\%2FBF01491914?LI=true}{\$39.95}~
\hypertarget{auto-label-caption-838202}{}
\hypertarget{auto-label-caption-786477}{}%
}}
\end{center}
\end{figure}\selectlanguage{ngerman}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/Hawking/Hawking}
\caption{{\hypertarget{auto-label-caption-202031}{}
\textbf{Particle creation by black holes}~\textbf{(1975)}
\hypertarget{auto-label-caption-400649}{}
\emph{by}~\emph{Stephen Hawking}
\hypertarget{auto-label-caption-789672}{}
This article uncovered that quantum mechanical effects cause black holes
to create and emit particles as if they were hot bodies.
(\emph{\protect\cite{Hawking_1975}})
\hypertarget{auto-label-caption-439481}{}
\textbf{Citations}: 9054
\hypertarget{auto-label-caption-554895}{}
\textbf{Cost}:
\href{http://link.springer.com/article/10.1007/BF02345020}{\$39.95}~
\hypertarget{auto-label-caption-378489}{}
\hypertarget{auto-label-caption-798679}{}%
}}
\end{center}
\end{figure}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/watts/watts}
\caption{{\hypertarget{auto-label-caption-737682}{}
\textbf{Collective dynamics of `small-world' networks (1998)}
\hypertarget{auto-label-caption-566896}{}
\emph{by Duncan J. Watts and Steven H. Strogatz}
\hypertarget{auto-label-caption-689149}{}
One of the foundational papers of modern network science, this paper
proposed a model of random graph generation which produces networks with
small-world properties, i.e., a graph in which most nodes are not
neighbors of one another, but most nodes can be reached from every other
node by a small number of hops or steps. \emph{}
(\emph{\protect\cite{Watts_1998}})\emph{~}
\hypertarget{auto-label-caption-152402}{}
\textbf{Citations}: 29942
\hypertarget{auto-label-caption-980740}{}
\textbf{Cost}:~\href{http://www.nature.com/nature/journal/v393/n6684/full/393440a0.html}{\$32.00}%
}\selectlanguage{english}}
\end{center}
\end{figure}\selectlanguage{english}
\begin{figure}[h!]
\begin{center}
\includegraphics[width=0.70\columnwidth]{figures/Screen-Shot-2016-08-26-at-4.56.01-PM/Screen-Shot-2016-08-26-at-4.56.01-PM}
\caption{{\hypertarget{auto-label-caption-494636}{}
\textbf{Does the Inertia of a Body Depend upon its Energy-Content}
\emph{by Albert Einstein}
Einstein's famous equation~\(E=mc^2\)~was developed in this
paper. The theory behind Einstein's equation -- ~perhaps one of the most
famous equations in physics-- is paywalled. (\protect\cite{Einstein_2005})
Citation: 804
Cost:~\href{http://onlinelibrary.wiley.com/doi/10.1002/andp.200590007/pdf}{\$38.00}
\hypertarget{auto-label-caption-138877}{}%
}}
\end{center}
\end{figure}

\emph{note:} this article was inspired by this
tweet:~\url{https://twitter.com/McDawg/status/767297109045501952}

\selectlanguage{english}
\FloatBarrier
\end{CJK}\end{document}

